Author's response to reviews

Title: Feasibility and costs of water fluoridation in remote Australian Aboriginal communities

Authors:

   Jonathon P Ehsani (jpehsani@hotmail.com)
   Ross Bailie (Ross.Bailie@menzies.edu.au)

Version: 6  Date: 7 June 2007

Author's response to reviews: see over
Dear BioMed Central Editorial Production Team,

Thank you for the acceptance of our paper (MS: 1979928173123119) on the feasibility and costs of water fluoridation in remote Australian Aboriginal communities.

As requested, the following changes have been made to the document:

The paragraph on authors' contributions has been removed from the title page.

The vertical lines in tables have been removed.

Sincerely

Jonathon P. Ehsani and Ross Bailie